
-----------------------------------------------------------------------------
          Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
       Mon Apr 13 16:19:32 GMT 2015
Year: 15  Doy: 103
Observer: berkey
WEATHER COMMENT: berkey: Mon Apr 13 16:19:33 GMT 2015
temp 37f, wind 5mph from the se, cirrus in the eastern half of the sky western have looks clear/hazy with the temp inversi
on capping the vog at about 12,000ft.
___end___
GENERAL OBSERVATORY COMMENT BY berkey: Mon Apr 13 16:37:23 GMT 2015
Container with new dome should get here in the next hour or so.
___end___
Mon Apr 13 16:39:41 GMT 2015 COMP Start Patrol
GENERAL OBSERVATORY COMMENT BY berkey: Mon Apr 13 17:50:35 GMT 2015
After a few years of waiting the new domes is here.  Well few hundred feet NE of this spot but its on the mountain :)
___end___
Mon Apr 13 18:04:34 GMT 2015 CoMP Paused for clouds
GENERAL OBSERVATORY COMMENT BY berkey: Mon Apr 13 18:10:54 GMT 2015
ulua cant mount mloserver.  Nots sure why (yet)
___end___
Mon Apr 13 18:29:32 GMT 2015 CoMP Restarted from pause
Mon Apr 13 18:32:05 GMT 2015 CoMP Paused for clouds
Mon Apr 13 18:34:31 GMT 2015 CoMP Restarted from pause
Mon Apr 13 18:35:13 GMT 2015 KCOR Start Synoptic Patrol
Mon Apr 13 19:17:47 GMT 2015 KCOR End Patrol
Mon Apr 13 19:18:09 GMT 2015 CoMP Paused for clouds
Mon Apr 13 19:25:53 GMT 2015 CoMP Restarted from pause
Mon Apr 13 19:27:09 GMT 2015 CoMP Paused for clouds
Mon Apr 13 19:39:11 GMT 2015 KCOR Start Synoptic Patrol
Mon Apr 13 19:39:17 GMT 2015 CoMP Restarted from pause
Mon Apr 13 19:50:07 GMT 2015 KCOR End Patrol
Mon Apr 13 19:50:39 GMT 2015 CoMP Paused for clouds
Mon Apr 13 19:52:51 GMT 2015 KCOR Start Synoptic Patrol
Mon Apr 13 19:52:54 GMT 2015 CoMP Restarted from pause
GENERAL OBSERVATORY COMMENT BY berkey: Mon Apr 13 19:54:07 GMT 2015
Moved DHCP from the server to the router. 
___end___
Mon Apr 13 19:56:27 GMT 2015 KCOR End Patrol
Mon Apr 13 19:57:09 GMT 2015 CoMP Paused for clouds
Mon Apr 13 20:03:34 GMT 2015 CoMP Restarted from pause
Mon Apr 13 20:05:09 GMT 2015 CoMP Paused for clouds
GENERAL OBSERVATORY COMMENT BY berkey: Mon Apr 13 20:04:09 GMT 2015
We have a "Dirac comb" of cirrus clouds today.  Clear Cloudy Clear Cloudy.  Its hard to keep up.
___end___
Mon Apr 13 20:38:26 GMT 2015 CoMP Restarted from pause
Mon Apr 13 20:38:26 GMT 2015 COMP End Patrol
Mon Apr 13 20:47:44 GMT 2015 COMP Start Patrol



GENERAL OBSERVATORY COMMENT BY berkey: Mon Apr 13 20:47:43 GMT 2015
Adjusted dome.  SGS realignment was very very small today.
___end___
Mon Apr 13 20:52:54 GMT 2015 KCOR Start Synoptic Patrol
Mon Apr 13 20:53:49 GMT 2015 KCOR End Patrol
Mon Apr 13 20:53:50 GMT 2015 KCOR Start Calibration script: c:\kcor\mlso-calibration22deg-20150323.ini
Mon Apr 13 21:10:57 GMT 2015 KCOR End Calibration Script
Mon Apr 13 21:11:13 GMT 2015 KCOR Start Synoptic Patrol
Mon Apr 13 21:11:14 GMT 2015 KCOR Start Synoptic Patrol
Mon Apr 13 23:25:56 GMT 2015 KCOR End Patrol
Mon Apr 13 23:26:09 GMT 2015 CoMP Paused for clouds
Tue Apr 14 00:01:20 GMT 2015 CoMP Restarted from pause
Tue Apr 14 00:01:21 GMT 2015 COMP End Patrol
GENERAL OBSERVATORY COMMENT BY berkey: Tue Apr 14 00:05:05 GMT 2015
Main dome closed for clouds to the NW
___end___
GENERAL OBSERVATORY COMMENT BY berkey: Tue Apr 14 00:22:10 GMT 2015
now closing pspt dome.
___end___


